---
title: "Intro to HPC Bootcamp 2025"
description: "Homepage for Sam Foreman's Intro to HPC Bootcamp 2025 Project"
categories:
  - ai4science
  - hpc
  - llm
toc: true
date: 2025-07-15
date-modified: last-modified
# modified: 2024-10-11
citation:
   author: Sam Foreman
   type: webpage
   title: "Intro to HPC Bootcamp 2025"
   url: https://samforeman.me/projects/hpc-bootcamp-2025
open-graph:
  title: "Intro to HPC Bootcamp 2025"
  description: "Homepage for Sam Foreman's Intro to HPC Bootcamp 2025 Project"
  # image: "./assets/thumbnail.png"
twitter-card:
  title: "Intro to HPC Bootcamp 2025"
  description: "Homepage for Sam Foreman's Intro to HPC Bootcamp 2025 Project"
  site: "saforem2"
  creator: "saforem2"
  # image: "./assets/thumbnail.png"

---

- [00-intro-AI-hpc/](./00-intro-AI-hpc/index.qmd)
  - [0-compute-systems](./00-intro-AI-hpc/0-compute-systems/index.qmd)
  - [1-shared-resources](./00-intro-AI-HPC/1-shared-resources/index.qmd)
  - [2-jupyter-notebooks](./00-intro-AI-hpc/2-jupyter-notebooks/index.qmd)
  - [3-homework](./00-intro-AI-hpc/3-homework/index.qmd)
  - [4-alcf](./00-intro-AI-hpc/4-nersc/index.qmd)
  - [5-mcmc-example](./00-intro-AI-hpc/5-mcmc-example/index.qmd)
  - [6-linear-regression](./00-intro-AI-hpc/6-linear-regression/index.qmd)
- [01-neural-networks/](../01-neural-networks/index.qmd)
  - [0-intro](../01-neural-networks/0-intro/index.qmd)
  - [1-mnist](../01-neural-networks/1-mnist/index.qmd)
  - [2-advanced](../01-neural-networks/2-advanced/index.qmd)
  - [3-conv-nets](../01-neural-networks/3-conv-nets/index.qmd)
  - [4-representation-learning](./4-representation-learning/index.qmd)
- [02-llms](./02-llms/index.qmd)
  - [0-intro-to-llms](./02-llms/0-intro-to-llms/index.qmd)
